

# Temi di discussione

(Working papers)

The topology of the interbank market: developments in Italy since 1990

by Carmela lazzetta and Michele Manna





# Temi di discussione

(Working papers)

The topology of the interbank market: developments in Italy since 1990

by Carmela lazzetta and Michele Manna

Number 711 - May 2009

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Patrizio Pagano, Alfonso Rosolia, Ugo Albertazzi, Andrea Neri, Giulio Nicoletti, Paolo Pinotti, Enrico Sette, Marco Taboga, Pietro Tommasino, Fabrizio Venditti.

Editorial Assistants: ROBERTO MARANO, NICOLETTA OLIVANTI.

## THE TOPOLOGY OF THE INTERBANK MARKET: DEVELOPMENTS IN ITALY SINCE 1990

### by Carmela Iazzetta<sup>\*</sup> and Michele Manna<sup>\*</sup>

#### Abstract

When a bank defaults or stops trading in the interbank market, both a liquidity shortage in the market itself and mounting trading losses should be anticipated. To gain more insight into the way a liquidity crisis spreads, we apply network topology techniques to monthly data on deposits exchanged by Italian banks, from 1990 to 2008. Our research yields three main results: first, only a few banks are today pivotal in the redistribution of liquidity across the system, while banks close to, but outside this core circle, weigh less than they used to; secondly, the halt in operations in a second set of banks may cut off some of their counterparts from the rest of the network, with increasingly less negligible effects; finally, only 2-3 banks out of the 10 we identify as most interconnected within the network are currently also among the top 10 banks by volume of traded deposits.

#### JEL CLASSIFICATION: D4, E5, G2.

Keywords: interbank market, topology, liquidity crisis.

#### Contents

0.0110110		
1. Introduction	5	
2. Some relevant literature		
3. Indicators of the topology of the interbank market		
4. Our dataset and some descriptive statistics		
5. Our results		
5.1 Measures of degree		
5.2 Measures of distance and degree distribution		
5.3 Measures of resilience		
5.4 Does topology matter after all		
6. Concluding remarks		
Appendices		
References		

<sup>\*</sup> Bank of Italy, Central Bank Operations Department. E-mail: carmela.iazzetta@bancaditalia.it and michele.manna@bancaditalia.it (corresponding author).



In 1736 Euler solved the riddle of the Seven Bridges of Königsberg: probably one of the first examples of applied topology.

#### **1.** Introduction<sup>1</sup>

A number of well-known features of the banking business – e.g. the maturity mismatch between banks' assets and liabilities and the sequential constraint banks have to fulfil when serving their customers – make it likely that a system-wide banking crisis materializes first and foremost as a sudden deterioration in trust. The most evident sign of this pattern is the bank run, which may strike both troubled banks as well as banks that would otherwise be sound.

In the market for interbank deposits trust plays a considerable role if one considers that deposits on the leading overnight (1-day) maturity are typically uncollateralized. Furthermore, because of this prevailing maturity, every day the lender faces the decision on whether to extend new credit or not, that is whether to renew its trust in the borrower. Hence, the widely held view that in good times the interbank market provides banks with a form of coinsurance against uncertain liquidity flows, while in bad times it acts as a catalyst in spreading the crisis. Furthermore, the potential for contagion of the crisis from a first bank to the system is magnified by the practice of major players to act simultaneously as lenders and borrowers in this market. As a result, the halt in the operations of one of them triggers two knock-on effects. The banks that have lent to the failed one will recover, at best, only part of their loan; in addition, the banks which used to borrow from the defaulted one will need to seek new funding parties, something which may not be straightforward at times of crisis. Thus, the default of a bank may cause both credit losses and a loss of liquidity for the market as a whole, two patterns also referred to as the *exposure contagion channel* and the credit line contagion channel.

<sup>&</sup>lt;sup>1</sup> We wish to thank Roberto Violi, Paolo Emilio Mistrulli, the participants at an internal seminar in the Bank of Italy and two anonymous referees. Cosma Onorio Gelsomino and Emerico Antonio Zautzik encouraged us first to carry out and then complete the research. We are especially indebted to Gerardo Palazzo, who helped us a lot in developing the programming routines, and Alice Chambers, whose accurate editorial work improved the text. Of course, all remaining errors are ours. While the research is the outcome of a joint effort by both authors, Carmela Iazzetta focused in particular on sections 4, 5.1 and 5.2, and Michele Manna on sections 2, 3, 5.3 and 5.4. The views expressed in this paper are those of the authors and do not necessarily reflect those of the Bank of Italy.

In this paper we examine the structure of Italy's domestic interbank market from January 1990 to June 2008 by means of network topology concepts and measures. Two questions guide our research. First, what is the locus of the banks in the network and which are most central to it? Put differently, which banks play a more important role in the redistribution of liquidity for the system (globally or locally) so that their demise would have the largest knock-on effect. Secondly, has Italy's interbank market evolved into a network that is more or less resilient to shocks as a result of the massive consolidation process the country banking system underwent during the last two decades?

Answers to these questions may provide insight into the relationship between developments in the structure of the banking system and the likelihood of a system-wide liquidity crisis. If, say, data pointed to an increased likelihood of contagion, then there would be an argument in favour of strengthening the banks' liquidity buffers across the board. Likewise, a better understanding of the contribution of each bank to the systemic liquidity risk could help to tailor its own liquidity buffers. These insights may be especially valuable when unfortunately it is no longer time for crisis scenarios since actual crisis management is at work. In these circumstances, the identification of the most interconnected banks becomes a key tool in targeting rescue packages and, say, the decision to grant Emergency Liquidity Assistance (ELA).<sup>2</sup>

The pursuit of these goals requires a methodology that works at the level of individual institutions and a dataset on bilateral exposures between banks. As to our methodology, network topology does not explain why a bank is first hit by an idiosyncratic shock. However, and crucially given our objectives, it will shed light on how a shock spreads across a network whose elements are internally connected, even more so when most of the links happen to be indirect, as is often the case in interbank markets (and the Italian one is no exception). Likewise, network topology provides the tools to identify the banks which may act as a trigger point in system-wide liquidity crises or, conversely, as a shock-absorber. Introducing some basic terminology, we will refer to the banks as the *nodes* and their interbank deposits as the *links* of the network.

The notion of shock spreading naturally brings to mind that of banks acting as global hub for the system. Their identification would be a relatively straightforward exercise - and accordingly network topology would make a less genuine contribution - if being interconnected and being large were mostly overlapping concepts. In fact, this statement does not hold true even when banks' size is measured narrowly on the interbank business: our results suggest that in 2007 in Italy only 2-3 out of the top 10 banks by volume of deposits were also among the top 10 by centrality in the network. Hence, the scope for using a methodology specifically geared towards the concepts of centrality and interconnectedness. A further important value-added of topology is its capacity to scan the network to search for "local hubs", that is banks whose day-to-day functioning is critical to some other banks, even though not necessarily for the system at large. Think of a bank p, which trades deposits with the rest of the banking system only through a bank q. If q stops operating - say because of a business continuity incident – then p would be cut off from the rest of system. Of course, bank p could look for a new counterparty, but this is not likely to be

<sup>&</sup>lt;sup>2</sup> In a speech delivered on 22 August 2008, the Chairman of the U.S. Federal Reserve, Ben Bernanke, outlined the rationale for the rescue of Bear Stearns: "Although not an extraordinarily large company by many metrics, Bear Stearns was deeply involved in a number of critical markets, including (as I have noted) markets for short-term secured funding".

achieved swiftly. At the very least, potential new counterparties would take their time to assess the creditworthiness of *p* itself.

Turning to the dataset, this should exhibit some defining features in order for network topology applications to add meaningful value to the research. If nand *m* denote the network's number of nodes and direct links, we would like: (i) *n* to be reasonably large; (ii) the ratio  $m/[n \times (n-1)]$  of active to potential links to be small, (iii) the network to be internally connected, so that for each pair of banks p and q it is always possible to shuttle funds between them, and (iv) the existing active links to be relatively stable over time. Conditions (ii) and (iii) imply that all pairs of nodes in the network are linked but in most cases the link is only indirect, condition (i) argues in favour of a developed methodology and reliable algorithms, while (iv) controls for the risk of random structures in the network and thus an unreliable identification of global and local hubs. Features (i)-(iv) are all but peculiar to the network topology of the interbank market. Albert and Barabási (2002) report some basic statistics of 17 networks studied in the literature (from math to words occurrence and power grids); for the median network, n was close to 22,000 and the ratio m/n was 0.02%, which becomes 1.0% when n<2,000.

Our main dataset is compiled using data on the month-end stock of outstanding deposits exchanged between banks located in Italy. The size of the resulting network is in the order of one thousand, its ratio m/n is in the range 1-1.5%, it is internally connected in each of the 222 months of our sample and, finally, its active links display a high degree of persistency. These elements meet the aforementioned desirable features and bode well for an application of network topology. Incidentally, this suitability seems to be found in a number of interbank markets also outside Italy, which may explain the stream of recent papers in this field.

Anticipating the gist of our results, we observe a number of changes in the topology of Italy's interbank network. Since 1990, fewer banks really act as global hubs within the network, while the gap between these top-notch market players and those one notch below has widened. At the periphery of the network, we observe a thinner structure with considerably less direct links. As a result, smaller players need more intermediaries to reach out to the rest of the system; at the same time the number of local hubs, in the sense defined above, rose strongly. All in all, this could be mixed news for policy makers. On the one hand, having to deal with less and more easily identifiable major players eases the task of targeting the support measures aimed at preventing an eventual system meltdown. On the other hand, the thinner structure at the fringes of the network means that even when major players are up and running, the circulation of funds throughout the whole network could be less than smooth and some pockets of the system could be left out. Hence, the need for the policy maker to keep a close eye not just on the largest players but also on a number of banks with more local yet relevant connections. One practical upshot could be the need to enhance measures for preventing business continuity incidents, including in relatively small banks.

What we do not observe directly is any major break in the interbank network after the onset of the current market turmoil. This should come only as a relative surprise having noted that in mid-2008 the total amount of outstanding interbank deposits was still roughly in line with pre-crisis levels. Nor, is there any sign yet of a break in statistics on the density of the network, on its inner correlation and on persistency in the bilateral relationships. All these patterns stand in sharp contrast with developments in the liquidity of markets, if only one observes that from June 2007 to September 2008 the volumes exchanged in the e-MID money market trading platform have shrunk to half. Taken together, these findings suggest that the role of the market in the redistribution of liquidity was compressed in favour of more bilateral and behind-the-scenes activity. This hypothesis suggests an agenda for future research, to which we will return in section 6.

The rest of the paper is organized as follows: section 2 reviews some relevant literature; section 3 sets out the algebra of a number of network topological measures; section 4 describes our dataset and introduces some descriptive statistics; section 5 presents our main results and section 6 concludes.

#### 2. Some relevant literature

Our research draws on a number of strands of literature. At a more theoretical level, following the work by Kindleberger (1978) and Bryant (1980), Diamond and Dybvig (1983) helped devise a model for bank runs due to a financial shock in a context of limited information, which leads to self-fulfilling prophecies. This is often referred to as the panic-based view on bank runs (Iyer and Peydró-Alcalde, 2005). The Diamond and Dybvig model has been expanded in several directions, e.g. by removing the constraint of limited information as a precondition for the run (Donaldson, 1992). It was up to Rochet and Tirole (1996) to develop a formal model of system-wide crises with n banks enjoying at 'time 0' different endowments, deviating from the single representative bank of previous papers. An alternative non sun-spot view regards financial crises as an inherent element of the business cycle, which results from asymmetric information between the banks and their customers (Jacklin and Bhattacharya, 1988).

Allen and Gale (2000) recognize three types of banking networks: (i) the complete structure, where each bank is linked to all other banks; (ii) the incomplete structure, where some of the links are only indirect; and (iii) the disconnected structure, where at least one pair of banks is not linked (either directly or indirectly). Their basic argument is that because of transaction and information costs, banks may refrain from acquiring claims on 'more distant' banks. However, while many banks deal directly with only a limited number of counterparties in the interbank market, the system as a whole is generally fully connected. The taxonomy defined by (i) to (iii) is relevant in policy terms because complete structures are usually thought to be more prone to contagion, but the consequences of the contagion may be less harsh. A more extreme situation is the money centre model of Freixas, Parigi and Rochet (2000), where the only existing links in the network are those to/from the money centre bank.

Eisenberg and Noe (2001) set out, using relatively straightforward algebra, the conditions for the existence and uniqueness of a clearing vector for a complex financial system.<sup>3</sup> Their work underlies a number of recent papers simulating the spreading of losses via interbank positions: basically, the simulation entails the assumption of the demise of a first bank bringing about losses to its lenders, which in turn could default, and so on and so forth. This

<sup>&</sup>lt;sup>3</sup> These authors impose two conditions to define a clearing payment vector: limited liability (total payments made by a node must never exceed the cash flow available to the node) and absolute priority (stockholders in the node receive no value until the node is able to completely pay off all of its outstanding liabilities).

strand of literature includes Sheldon and Maurer (1998), Furfine (2003), Blåvarg and Nimander (2002), Wells (2002), Cifuentes (2004), Upper and Worms (2004), van Lelyveld and Liedorp (2006) and Mistrulli (2007) who work on U.S., Swedish, Chilean, German, Dutch and Italian data. Broadly speaking, these papers end on a positive note, since they judge the default of a single bank, no matter how large, unlikely to cause the failure of large sections of the banking system. A more critical view is offered by Brusco and Castiglionesi (2007), who link the finding of limited potential for system-wide failures to the assumption of a fixed structure in the interbank claims. In their view, the final impact could be much larger if the prices and the balance sheets of the banks were allowed to vary endogenously along with the transmission of the shock. Degryse and Nguyen (2007) add a time dimension to the simulated sequence of defaults. Their main result is that the way interbank linkages may be conducive to contagion does change over time, occasionally it does so swiftly after a major shock such as September 11. A further specific note goes to Müller (2003), who applied Swiss data to expand the basic simulation set-up to embody liquidity rationing along with the 'more standard' credit losses.

To various degrees, all these works include some (implicit) reference to the concept of interbank network, where the size and direction of links are crucial to assessing the extent to which the network is prone to contagion. Notwithstanding recent developments, the specialized literature on market topology is still relatively limited. Formal measures of topology can be found in Müller (2003 and 2006), Boss et al. (2004), Soramäki et al. (2006), Cajueiro and Tabak (2007), Iori et al. (2007), von Peter (2007), Becher et al. (2008), Bech and Atalay (2008) and Nier et al. (2008). The last paper offers good hooks to the main results of this literature. Higher levels of the ratio between active and potential links in a network (higher density) are thought to be conducive to swifter shock transmission, although the relationship may happen to be non linear.<sup>4</sup> For a given level of density, higher concentration in the banking systems leads to more vulnerability to system-wide breakdowns. Finally, it is not always the case that tiered banking systems are more fragile than relatively homogeneous systems, since much depends on the type of links run by the banks at the centre of the network.

This stream of applied research has benefited from the increase in and greater access to enhanced computing capabilities. In addition, and perhaps more recently, a trigger for a number of papers is the more widespread understanding that some intermediaries may be "too-interconnected-to-fail" (jointly or alternatively to being "too-big") and thus ultimately the quest for understanding first, and reducing later, systemic risk in financial markets. The relative novelty of network topology in finance should not mislead anyone. In other fields this methodology is well-established and prides itself on an honourable record.<sup>5</sup> Possible references include Albert and Barabási (2002) as an example of topology's application to physics, Watts (2003) to sociology, Bollobás (1998) to mathematics and Albert, Jeong and Barabási (2000) to the internet and World Wide Web. By way of example of an oft-cited result of the literature, many networks are proved to have a low density but nonetheless

<sup>&</sup>lt;sup>4</sup> The sign of the relationship between connectivity and contagion may cease to be ambiguous when the ratio is in the region of 20% to 50%. In practice, these levels are seldom found in real-world networks (Albert and Barabási, 2002). A counter-example with a density of 88% is in Becher *et al.* (2008), who, however, examine a network of only 15 nodes.

<sup>&</sup>lt;sup>5</sup> In his survey Newman (2003) cites an impressive number of 429 papers, of which 18 are from the 1970s, 10 from the 1960s, 7 from the 1950s, 2 from the 1930s and 1 from the 1910s.

display the so-called small world pattern, that is most pairs of nodes are not directly linked but they keep being connected through a limited number of intermediaries, often not larger than six in sociological applications. Hence, the well-known law of 'six degrees of separation' (Watts and Strogatz, 1998 and Newman, 2003).<sup>6</sup> Moreover, the consensus view is that most real-world networks are highly robust, in the sense that "while key components regularly malfunction, local failures rarely lead to loss of the global information-carrying ability of the network" (Albert and Barabási, 2002).<sup>7</sup>

The resilience of a network to random failures in its nodes is closely linked to the distribution of the number of links across the nodes. When this distribution is "scale-free" – roughly speaking, if *n* nodes in a network each hold *m* links, then (approximately)  $\alpha \times n$  nodes will hold  $\beta \times m$  links where the parameters  $\alpha$  and  $\beta$  (both < 1) do not change with the starting point *n* – the literature suggests that the network is highly resilient to random failures. Unfortunately, this 'virtue' of scale free networks becomes a 'vice' when the failed nodes are the result of a well-targeted attack rather than being the result of some random process. In that case, even a low failure rate may seriously hamper the network.

Even a brief survey on topology cannot omit to deal with terminological and data issues. We introduced above the terms 'node' and 'link', which we took from the literature on networks in information technology, and more jargon will ensue. Readers should be aware that they may encounter different sets of terms in different contexts: the node becomes a site in physics and an actor in sociology with vertex as a further alternative; again, the link becomes a bond and a tie in physics and sociology respectively. Even the word network itself may be replaced and mathematicians more commonly speak of a graph.

Lack of suitable (bilateral) data often hinders network topology applications in finance. To our knowledge, the best an author can rely on is, as in Mistrulli (2007), a dataset featuring actual gross outstanding deposits reported monthly by each resident bank (or, when datasets at a higher frequency are available, usually coverage is less than full). This is already a great database if, say, Upper and Worms (2004) had to construct their own dataset from a set of plausible assumptions. Less data teething problems occur in analyses of (intraday) payment system flows. The earlier dates of the works by Angelini, Maresca and Russo (1996), and Koponen and Soramäki (1998), who used intraday data for Italy and Finland respectively, should therefore come only as a partial surprise. More recently, Devriese and Mitchell (2005) provided a primer of applied empirical research on Securities Settlement System data, where the financial intermediary is potentially subject to two forms of illiquidity, that of cash and securities. Clearly, the trade-off in using daily data is the associated computational burden, which limits the width of the sample, bearing in mind that network topology is already a computationally demanding technique (a fortiori, this holds true with respect to intraday observations).

<sup>&</sup>lt;sup>6</sup> The law is often associated with the work of Milgram (1967), although the concept had already received attention in earlier times. In his 1909 Nobel Prize speech, Guglielmo Marconi noted that roughly six transmission stations would be needed to cover the inhabited areas on the earth (in his computations, the exact number was 5.83).

<sup>&</sup>lt;sup>7</sup> Albert, Jeong and Barabási (2000) prove that a failure of five percent of randomly chosen nodes, a very high failure rate by most standards, has nearly no impact on communications within the network.

#### 3. Indicators of the topology of the interbank market

We consider an interbank market with *n* banks; this number is the size of the network. Each bank represents a <u>node</u>. The outstanding deposit lent by bank *p* to bank *q* is worth  $w_{p,q} \ge 0$ ; the corresponding amount lent by *q* to *p* is  $w_{q,p}$ . Since we deal with data on gross outstanding bilateral exposures,  $w_{p,q}$  and  $w_{q,p}$  may be jointly positive.

There is a direct <u>link</u> between p and q if  $w_{p,q} + w_{q,p} > 0$ . If  $w_{p,q} = w_{q,p} = 0$ , nodes p and q are not directly linked. Below, the concept of 'directness' in a link will become clear when we introduce the distance between two nodes. We define explicitly the link corresponding to  $p \equiv q$ ; the weight  $w_{p,p}$  is obviously nought. This enables us to compile an  $n \times n$  matrix W, known in the literature as the weighted connectivity matrix, whose generic element of cell p,q is  $w_{p,q}$ .<sup>8</sup> From W, we derive two  $n \times n$  matrices. The 'adjacency' matrix S flags whether the direct link between two nodes exists or not, irrespective of its weight. Its generic element  $s_{p,q}$  is:

$$\begin{bmatrix} 1 \end{bmatrix} \quad s_{p,q} = 1 \qquad \text{if} \quad w_{p,q} > 0 \\ 0 \qquad \text{otherwise} \end{bmatrix}$$

In addition, we refer to a symmetric matrix B which highlights whether a link between nodes p and q exists, irrespective of the direction:

$$\begin{bmatrix} 2 \end{bmatrix} \qquad b_{p,q} = 1 \qquad \text{if} \qquad w_{p,q} + w_{q,p} > 0 \\ 0 \qquad \text{otherwise} \end{bmatrix}$$

We also use an  $n \times 1$  vector  $i_g$ , which takes value 1 in the g-th cell and 0 otherwise, and an  $n \times 1$  vector u, filled in with all 1's. The symbol "T" denotes the transpose of a matrix, e.g. S<sup>T</sup> transpose of S. Finally, we occasionally make use of a function 1(X=m), which applied to the matrix or vector X returns the number of cells in X which take value m.<sup>9</sup>

#### *(i)* Degree

The degree is the number of active links originating in node g or terminating at it. We define an out-degree and an in-degree which, when applied to the interbank market, correspond to the number of banks to which g lends to and from which it borrows. In many instances, it is also useful to count the total number of banks to which g is connected, irrespective of the direction of the link.

- [3]  $k_g^{OUT} = i_g^T S u \in [0, n-1]$  out-degree of node g
- [4]  $k_g^{IN} = i_g^T S^T u \in [0, n-1]$  in-degree of node g
- [5]  $k_g = i_g^T B u \in [1, n-1]$  degree of node g

<sup>&</sup>lt;sup>8</sup> In W we report the lending bank row-wise and the borrowing one column-wise. To count averages, the sum is divided by  $n \times (n-1)$ , the maximum number of links which may be activated.

<sup>&</sup>lt;sup>9</sup> We do not add a time prefix in our formulae, although our statistics are in principle time variant, starting from the number n of banks in the network at each end month. This will ease notation, hopefully without introducing too much confusion because the meaning ought to be clear from the context.

Next, we compile the  $n \times 1$  vector k whose g-th element is  $k_g$  and its akin vectors  $k^{OUT}$  and  $k^{IN}$ . The ratio of the sum of the in-degrees across all g's, or equivalently of the sum of the out-degrees, divided by  $n \times (n-1)$  yields the density (aka connectivity), a basic statistics of any network:

[6] 
$$p = \frac{u^T S u}{n(n-1)} \in [1/n, 1]$$
 density (connectivity)

Subsequently, we derive the degree distribution function, by counting how many nodes are of degree 1, 2, ..., m, ..., n-1:

[7] 
$$P_m = 1$$
 (k = m) degree distribution function

This function helps to differentiate between two broad classes of networks: the exponential network where most nodes have approximately the same number of links, and the scale-free network, where the majority of nodes hold only a few links but a limited number of nodes boasts a much larger number of links.

Two indices of correlation may yield insights into the structure of the network. The so-called degree correlation gauges the relationship between inand out-degrees:

[8] 
$$\rho^{K} = \text{correlation}\left(k^{OUT}, k^{IN}\right)$$
 degree correlation

while the degree of assortativeness measures the relationship between the degree of each bank g and the average degree of the banks directly linked to g itself. To this end, we follow a two-step procedure

[9a]  $\hat{k}_{g} = \frac{\hat{i}_{g}^{T}BBu}{k_{g}}$  average degree of nodes linked to g

[9b]  $\rho^{A} = \text{correlation}(\mathbf{k}, \hat{\mathbf{k}})$  degree of assortativeness

where the structure of the two  $n \times 1$  vectors k and  $\hat{k}$  should by now be obvious. If  $\rho^A > 0$  the network is said to be <u>assortative</u>, i.e. nodes usually mate with nodes of similar degree; if  $\rho^A < 0$  the network is disassortative.

#### *(ii)* Valued degree

This is a set of measures mirroring those of the degree except that they embody the weight of the links. Related formulae are in appendix 1.

#### (iii) Centrality

Measures of 'centrality' describe the position of the node within the network. To this end, a key concept is the <u>distance</u> between the nodes p and q,  $d_{p,q}$ , defined as the shortest path between them:

[10] 
$$d_{p,q} = m < n$$
 if  $\exists m-1 \text{ nodes } a', a'', ..., a^{[m-1]} \text{ s.t. } b_{p,a'} = b_{a',a''} = ... = b_{a[m-1],q} = 1$ 

Nodes *p* and *q* are linked directly if  $d_{p,q} = 1$ ; they are linked indirectly if  $d_{p,q}$  is finite and greater than one; otherwise they are not linked. If any pair of nodes

in the network is mutually linked, then the network is internally connected. Finally, a bank g is said to be geodesic as regards banks p and q if it stands on the shortest route between them. Algebraically, g is geodesic to p and q if

[11a]  $d_{p,q} = d_{p,g} + d_{g,q}$   $g, p, q \in [1, n]$ 

against the alternative of

[11b]  $d_{p,q} < d_{p,g} + d_{g,q}$   $g, p, q \in [1, n]$ 

Further details and the algebra of additional measures of centrality, inter alia the weighted distance and the ergodicity of a node, are in appendix 2.

#### (iv) Resilience

We adopt two approaches to gauge the resilience of the interbank network: (a) one node at a time is removed and the knock-on effects on other banks in the network are worked out (Furfine, 2003); (b) a less common approach that aims at identifying the banks with highest geodesic frequency (Müller, 2003). The two approaches work quite differently: (a) helps to understand how much thinner the web becomes when a bank defaults and its direct links disappear; (b) looks more at the indirect links as the backbone of the overall circulation of funds throughout the system.

Figure 1 shows three simple networks all of size 7, each populated by a bank g and six banks called '1' to '6'. In view of their small size, the three networks also have similar density, 29% to 38%. Differences arise when g is removed, so that its links (dotted in the figure) disappear: in the left-hand panel, only bank 1 is cut off from the rest of the system; in the centre panel this happens to banks 1, 2 and 3 while 4, 5 and 6 form a new network; finally, in the panel on the right-hand side each of the six surviving banks is isolated and deposits are no longer exchanged.



Dotted lines show links which are no longer active after the removal of bank (node) g.

Arguably, moving from left to right we observe more severe forms of disruption, even if the starting conditions are quite close to each other. Even more diverse assessments would be derived if the weights of the nodes were taken into consideration. Consider, for example, when bank '1' in the left-hand panel is smallish or conversely fairly large. The algebra of the two measures of resilience is set out in appendix 3.

#### 4. Our dataset and some descriptive statistics

As anticipated in the introduction, our main dataset is based on month-end gross unconsolidated bilateral interbank deposits reported by Italian banks to the Bank of Italy for supervisory purposes. In particular, we focus on the deposits exchanged with other resident banks. The time series runs from January 1990 to June 2008, for a total of 222 monthly data points. The number of banks covered in the dataset (the size n of the network) is 1,146 at the beginning of the sample, and 790 at the end.

A number of reasons may explain the dominance in the literature of recourse to data on unconsolidated deposits vs. data on banking groups' net exposures. In cross-border groups, fully centralized management is challenged by the obstacles in the flow of cash and securities across borders identified in BIS (2006). Within the domestic component of the groups, a preliminary data investigation showed that occasionally more banks run simultaneously a significant off-the-group interbank activity, hinting at structures of less than full centralised liquidity management. Moreover, the recourse to consolidated data is tantamount to assuming that intra-group credits enjoy seniority vis-à-vis credit claimed by banks outside the group, which is not the case (Mistrulli, 2008).

In the period under examination, the amount of interbank deposits traded by Italy's banking system grew considerably. In 1990, Italian banks reported deposits worth €85 billion, comprising €45 billion vis-à-vis other domestic banks and €40 billion of cross-border deposits (average of month-end stocks). Seventeen years later, the corresponding figures were €611, €433 and €179 billion.<sup>10</sup> This gives an average annual growth rate of 12.3% on total interbank deposits, which even adjusting for inflation remains high at 8.5%. The growth rate outpaced developments in the entire banking business: as a result, the weight of such deposits out of the total size of the banking system's balance sheet rose from 9-13% in the early 1990s to 17% in 2008.



<sup>&</sup>lt;sup>10</sup> The share of cross-border positions fell from half (40 billion out of 85) to less than one third (179 out of 611). No hasty conclusions should be drawn from this pattern on the integration of Italian banks in the euro money market, which is verified by the law of one price. If market integration is to be gauged from volumes, this should be based on a criterion of indifference *ceteris paribus* towards the country of residence of the counterpart, not on the sheer data on volumes (Manna, 2004).

Besides getting bigger, the interbank market also became more concentrated. The market share of the top-50 banks grew from 69% on average in 1990-1992 to 87% in 2005-2007; perhaps more tellingly, the deals where both the lending and the borrowing side was a top-50 bank surged from 47% to 78%. The largest players remain net borrowers from smaller ones, a recurring finding of studies on the microstructure of the interbank market. The gap, however, has almost closed: from 7 percentage points in the first three years of our sample, 26% of gross borrowing minus 19% of lending, to just 0.2 percentage points, (9.1% minus 8.9%).

Table 1

		Borrowing bank					
		Average 1990-1992			Average 2005-2007		
		Top 50	Other banks	Total	Тор 50	Other banks	Total
Lending bank	Тор 50	47.1	18.9	66.0	78.4	8.9	87.3
	Other banks	25.6	8.4	34.0	9.1	3.6	12.7
	Total	72.7	27.3	100.0	87.5	12.5	100.0
Overall market share of top-50 banks		of		69.3			87.4

Distributio	n of interban	k deposits ac	cording to	banks' rankin	g
(percent	tage values: depo	sits exchanged	between two re	esident banks)	

Each cell reports the share of deposits lent by the group of banks reported row-wise to the group of banks reported column-wise. The top 50 banks are identified from their ranking on total (lending + borrowing) deposits.

The process of market concentration developed along the more general consolidation of the Italian banking system. In June 2008 the five largest banking groups accounted for 61% of the total assets of Italy's banking system, up from 45% at end-1999 and 33% at end-1993. Corresponding shares for the top 5 individual banks were 22% in 1993 and 33% in 2008. The latter milder pattern likely reflects a process of consolidation which took place more often through acquisitions rather than mergers.



Figure 3

To disentangle the contribution of the system's consolidation onto the market enhanced concentration, we compiled an additional dataset covering only the outside-the-group interbank exposures. Having done so for six selected dates, the overall market share for the top 50 banks measured increased from 70.5% in January 1993 to 80.6% in June 2008, while the corresponding figures worked out on the main dataset (which covers both within- and outside-the-group deposits) are 70.5% and 83.3%. We gather that only one fourth of the observed increase in concentration may be directly related to the enhanced system's consolidation, supporting the view that the concentration mainly took place across and outside groups. In turn, this upholds the hypothesis that some banks are gaining ground as money centres for the whole banking system and not only within the group to which they are affiliated.

Table 2

### Interbank market concentration: the role of group consolidation

(percentage values)							
	Jan 1993	Feb 1996	March 1999	April 2002	May 2005	June 2008	
	Dataset including both within and outside group deposits						
Overall market	70.5	69.3	70.3	78.2	85.5	83.3	
share of top 50 banks	Dataset including only outside group deposits						
	70.5	70.1	71.3	74.0	79.6	80.6	

In annex 4 we describe additional checks on the comparability of our dataset with other datasets studied in the network topology literature, on the persistency of bilateral relationships in interbank trading and on the impact of changes in the banking population.<sup>11</sup>

<sup>&</sup>lt;sup>11</sup>As a rule, we exploit the whole dataset and the full time series in our applications. In two ad hoc exercises we focussed on six dates. Additionally, in computing a measure of resilience to shocks we omitted the "Banche di credito cooperative" (BCC), accounting for one-half of n of the main dataset but only one-tenth of the deposits in value. Both choices were dictated by sheer computational burden. For example, the exclusion of the BCCs reduced the number of triplets to be checked from 1 billion to "only" 120 million.

#### 5. Our results

#### 5.1 Measures of degree

The average Italian bank exchanged deposits with 10 counterparties in 2007, down from 17 in 1990; if this average is adjusted for the changing size of the network, the pattern is less regular although a downward trend emerges clearly anyway (figure 4). We obtain from the outset a picture of a relatively scanty network where only 1-1.5% of the links are active. The variability in individual data is relevant and the standard deviation is twice the mean for the out-degrees and three times for the in-degrees.

Figures 4a-b



Network degree: measures of average and standard deviation

To shed light on the patterns prevailing at a micro level, in the following we focus on the bank which each month is ranked first by number of counterparties (degree), as well as those on the 95<sup>th</sup> and 50<sup>th</sup> percentile (henceforth, the "top bank", the "95<sup>th</sup> pc bank" and the "50<sup>th</sup> pc bank"; occasionally, we will refer also to a "90<sup>th</sup> pc bank", whose meaning is apparent). Starting from the top bank, its in-degree fell from 841 in 1990 to 439 in 2008, its out-degree from 355 to 251. The top bank borrows from a higher number of counterparties than the one it lends to, in a ratio which decreased throughout our sample from 2.4 to 1.7. Again, this supports the view of an increasingly more even dealing behaviour by the largest banks.





The chart on the left shows the result of [6] and of its numerator, the former in %. The chart on the right shows the ratio between the standard deviation of the in-degree of all n banks and the average, and the corresponding measure for the out-degrees.

In figure 5 we flag two dates when patterns could have changed swiftly: end-December 1998, the day before the switch from lira to euro, and end-September 2001, the first month-end after the Twin Towers attacks. On both occasions, a simple eye-ball inspection suggests that the reaction of the monthly statistics was virtually muted and the same can be said of most of our statistics. Drawing from other studies which, using daily data, show that the properties of interbank networks did change on September 11 but the impact was short-lived<sup>12</sup>, an educated guess is that the reaction had disappeared already by the end of the month. This may explain why the reaction did not show up in our series (an alternative explanation could be that the Italian interbank market was not affected at all).

The in-degree of the 95<sup>th</sup> pc bank fell from 81 to 34, its out-degree from 68 to 32. This is a more marked decline than the one described above for the top bank. As a result, the ratio between such measures for the two banks increased from 10.4 to 12.9 for the in-degree and from 5.2 to 7.8 for the out-degree. The result is a weakening role by the 95<sup>th</sup> pc bank (approximately, ranked 40<sup>th</sup>). In a nutshell, if before we described the interbank market as a game effectively played by 50-odd banks, we start gathering evidence that within this club the gap between the top and the close-to-the-top banks has widened.



The chart on left shows the in- and out-degree worked out on the bank ranked at the 95<sup>th</sup> percentile according to these measures each month. The chart on the right shows the corresponding measures for the median bank.

The linear correlation between in- and out-degrees ranges between 0.8-0.9, while the corresponding weighted correlation is often lower and certainly bumpier. The two findings hint at a web of relationships where banks prefer to continue dealing with their usual counterparties, while being less concerned by the variability in the volume of associated deals. The network is also proved to be disassortative, suggesting that active banks frequently deal directly with smaller players, although less today then it used to be in the early 1990s.

 $<sup>^{12}</sup>$ Soramäki *et al.* (2006) note that the density of the network defined by interbank payments over the Fedwire Funds service in the U.S. dropped after the attacks from 3.0% to 2.6%, to shoot up again to over 3% already from the 14th to the 17th.



The chart on the left shows the linear and weighted correlation between in and out-degrees, using [8] and [A.4]. The chart on the right shows the degree of assortativeness, using [9b].

#### 5.2 Measures of distance and degree distribution

This analysis of the interbank market microstructure may be enriched using the degree distribution function. Much like a number of other networks, ours displays a smooth scale-free function with a large asymmetry on the left. This supports the view of our network as more robust to random failures than to targeted attacks. On average over the whole time span, 88 banks held five links, but only 25 banks held 10 links and 2 reached 50 links. That is, only a handful of Italian banks boasts a large number of counterparties in the interbank market; a similar result is in Boss *et al.* (2004) for the Austrian market. Figure 8b breaks down the function by subperiods (we cut the x-axis to 25 as most of the variability is on the lower degrees). Over time, the mode of the distribution shifts to the left and, on the vertical of 10 links, it moves down offering further evidence of a diminished role in the market of the relatively large players.



The chart on the left counts how many banks hold one link, two links .. on average over the whole sample. The chart on the right does the same with reference to four selected sub-periods. The circle highlights the area of shift to the left of the curves, along the vertical of 10 links.

Through this set of results we also derive the mass distance function. In the early 1990s, the distance was 2 in more than 80% of the pairs of banks and 3 in 10% of them. Proportions began to change in the late 1990s and, by June 2008, they had become respectively 54% and 43%. In total, at the latter date we counted 56% of pairs within 2 links and 99% within 3 links, reasonably close to the 41% and 95% obtained by Soramäki *et al.* (2006) for the US Fedwire.



The chart shows the percentage of pairs of banks whose distance is, on each month, 2 and 3.

The end result is an increase in the average path length, from 2.1 to 2.4. The eccentricity (maximum value) of the distance went up from 3-4 to 5: if the rule of thumb described by the law of 'six degrees of separation' is adapted to a network of relatively small size, this finding provides further evidence of a network that is well behaved as far as its topology goes.

If the average linear distance increases, the corresponding weighted measure does so much more rapidly. The drift away from the centre of the smallish banks located at the fringes of the network may not suffice to explain this finding, because of the associated low weight. Rather, we should more reasonably search for an explanation in the drift away of the large-but-not-top banks, whose weight is more relevant.



The chart on the left shows the average linear (unweighted) distance, result [A.6]. The chart on the right shows values of the median version of [A.6] and [A.9].

#### 5.3 Measures of resilience

A first approach to measure the resiliency of the network of interbank deposits insists on removing each bank one at a time and identifying which of them make, as a result of the removal, the network disconnected. The number of banks boasting this 'disconnecting skill' is on the rise: from 5 or less in the early 1990s to around 15 in recent years.

Figure 11



We argued above that in a disconnected network the split may be more or less severe. To work out a more precise measure of this intuition, first we identified the "disconnecting banks". Second, we simulated their removal and measured the area of the network, which remained cut off from the core (using [A.13]). Third, we ranked the derived measures in descending order and binned into a higher, medium and low third. Fourth and finally, we calculated the average for each of the three groups. The procedure was repeated for six dates. We already knew that the number of the "disconnecting banks" had increased. Through this procedure, we observe a heterogeneous impact on the network of their removal, across banks and over time. For the banks in the higher third, the measure of the disconnected area more than doubled comparing the results for January 1990 and June 1994 to those of later dates (from less than 2% to 4-5%). Conversely, the impact is more limited and notably there is no visible trend when we turn to the banks grouped in the medium and lower third. One interpretation of these results is that while "global hubs" have increased in role (when assessed against their potential for contagion), "local hubs" have increased in number.





Number of banks which, if removed, make the network disconnected

Figure 12

The fact that the demise of a bank is able to cut off a non-marginal portion of the network may depend on the size of the bank's interbank exposures (as lender and borrower) but also on the number of its direct counterparts, for given size. Above, we gathered evidence on the enhanced market concentration; here we examine the latter profile through measures of geodesic frequency. Having worked out the measure of such frequency for each bank and each month, the maximum monthly frequency is proven to fluctuate at around 60%, with no visible pattern. That is, the observed average reduced in density (number of links) described in figure 6a did not extend to the bank most central to the network.

The two results are reconciled by the pattern followed by the geodesic frequency of the 95<sup>th</sup> pc bank, which declined from 20-30% in the early 1990s to 7% in the most recent observations. The locus of this bank in the network has shifted away from the centre. A simple way of measuring this shift is through the ratios between the two frequencies now described, a ratio which went up from 3-4 in most of the 1990s to 6 in more recent years (figure 13b). At the same time, this shift was not the outcome of a parallel expansion across the whole network, as suggested by the convergence to one of the ratio between the geodesic frequency of the 95<sup>th</sup> pc and 90<sup>th</sup> pc banks.



The chart on the left shows the geodesic frequency of the "top bank", "95<sup>th</sup> pc bank" and "90<sup>th</sup> pc bank", calculated using [A.14]. The chart on the right shows two derived ratios.

Further evidence can be collected by counting the number of banks with geodesic frequency above a given threshold, set equal to one half and one third in the results shown in figure 14 (overleaf). Until 1997, 15 odd banks passed the threshold of one third and 6-8 that of one half; after a rapid drop around the turn of the millennium and some limited recovery, the corresponding statistics are respectively of 3-5 and 1 in 2008.



#### 5.4 Does topology matter after all?

By and large, so far we have tried to identify which banks are more pivotal in the transfer of deposits across the banking system (more central to the network). One possible conclusion is that in this context the use of topology helps identify the big banks, even if through a different and less common perspective. In fact, such a common wisdom interpretation appears to hold true only if we attribute a loose meaning to the adjective 'big'. Figure 15 shows the number of banks which are both in the top 50 list by geodesic frequency and in the top 50 list by total interbank deposits, along with the number of overlaps for the two corresponding top 10 lists. In the mid 1990s some 40 banks were in both top-50 lists, a number which fell to 30 in the mid 2000s. If we restrict the analysis to the top 10 lists, the number of 'dual' banks goes down from 6-7 banks to 2-3: the overlap halves and hits much lower values.





The chart shows the number of banks which are in both top 50 lists, by size and by geodesic frequency, and the corresponding number for the two top 10 lists.

Noticeably, in our results, the group of banks whose halt in business (their removal in our simulations) may render the network disconnected turns out to be a further distinct group, since it generally does not overlap with the banks with a high geodesic frequency, or with those running large volumes of interbank deposits. This suggests that while troubles with "disconnecting banks" are less likely to cause system-wide consequences, they may nonetheless have non-marginal consequences for some of their direct counterparts.

#### 6. Concluding remarks

In this paper we examine the structure of Italy's domestic interbank market, applying a network topology approach to a monthly series of stock data from January 1990 to June 2008. Our research set out to answer two major questions. First, how central are banks in the network and how has their positioning changed over time? Second, has the network become more or less resilient to shocks, also in the light of the parallel process of consolidation of the country banking system?

The country's interbank market has the following main topological features. First, it displays a low and decreasing density. If, in the early 1990s the average bank exchanged deposits with 1.5% of other resident banks, towards the end of the sample this percentage was close to 1.0%. Second, the network remains connected in each of the 222 months we covered. This means that it is always possible to find a route, on already existing links, to shuttle funds between any pair of banks. Third, the distance between the nodes – the number of intermediaries required to shuttle the funds – is gradually but steadily rising. We can think of an expanding network, with the important qualification that the speed of expansion is not homogeneous across the network. A number of findings show that the ring of nodes in the network, which stand just a notch outside the core – what we have nicknamed the close-to-the-top banks, whose behaviour can be epitomized by the 40th-odd bank by number of counterparties - is drifting away more rapidly. This makes the role of such banks in the market more similar to that of less important players while its gap from the core – which is less populated in 2008 than it used to be in the early 1990s - has widened. Fourth, the network is found to be disassortative, albeit increasingly less so, where this technical term signals that major players often deal directly also with smaller ones. Fifth, the estimated degree distribution function for Italy's domestic interbank market is highly smoothed, closely resembles the textbook scale-free type and, again, offers evidence of a diminishing role by the relatively large players.

A number of these findings, notably the low density, the interconnectedness and the scale-free distribution, suggest that our network is well behaved as far as its topology goes. Hence, inferring from the results of a rich and wellestablished empirical literature (in finance as well as in several other fields), one could, for example, view the interbank market in Italy as highly resilient to shocks that randomly hit individual banks.

The patterns observed in several of our statistics point to a transition, which is under way since roughly the turn of the millennium or just before it. The model, which prevailed until the early 1990s – whereby large banks used to be net borrowers from the rest of the system, and dealt directly with a number of small players at the fringes of the network, while the close-to-the-top banks were not so distant from the really top ones – is on the wane. At the same time, an alternative model of a tiered banking system, where an elite of wellidentified leading players is largely segmented from smaller players, appears to have not yet been fully accomplished. For example, while the declining degree of disassortativeness speaks in favour of this transition, we would wait for the degree to change sign (the system turning assortative) before calling the case closed.

A thinner and expanding network is also more subject to being frayed, especially at the edges. More precisely, it is possible to identify the banks whose halt in the operations, due to their default but also to less extreme events, such as a technical incident which temporarily knocks out their IT centre, cuts one or more of the other banks off from the rest of the system. The number of such "disconnecting" banks is clearly on the rise, from less than 5 in the early 1990s to almost 20 in more recent years. Because in most cases the size of the network becoming disconnected is relatively limited (less than one per cent), we may think of such banks as local hubs in the system.

At the same time, perhaps as a result of the already mentioned reduced role of an intermediate layer of banks which contributed to keep the network connected, the demise of one of the largest players would have twice as large an impact in terms of disruption to the circulation of funds than it had in the early 1990s.

Such a diverse set of results implies that any assessment of changes in the network shock resilience need to be qualified. We said earlier that fewer banks populate the very centre of the network and really act today as its global hubs. Hence, we are confronted with a smaller number of banks with a large potential for spreading a crisis throughout the system. However, if these banks are hit, the impact can be greater than it used to be. In addition, more banks can trigger disruptions in liquidity distributions, where this disruption may be small on aggregated terms but highly relevant for specific pockets of the banking system.

From the point of view of the policy maker, these developments could be mixed news overall. On the one hand, a more precise identification of the banks acting as global hubs in the process of redistribution of liquidity should ease the targeting of support measures to systematically relevant institutions. On the other hand, the potential for damage brought about by the demise of these banks has increased over time. Moreover, due to the rise in the number of local hubs, the fact that major market players are up and running is no longer a sufficient condition to ensure the smooth circulation of funds across the whole banking system.

Our results suggest that the overlap between the cluster of big banks and that of most interconnected ones in the interbank market is limited, even when banks' size is measured narrowly on the volume of its interbank deposits. This finding suggests that crisis management could gain from a deeper understanding of the structure of this market, as network topology may provide.

It is beyond the scope of our paper to examine how customer relationships are established in the interbank market and why ultimately bank p chooses to deal with bank q and not with g. As part of a broader data investigation there remains, however, the empirical finding of a high persistence in the bilateral relationships between banks. This suggests that the search for a new counterpart may not be fulfilled swiftly and is all but left to the invisible hand of an auctioneer. This finding is likely to hold even more when a crisis has broken out.

A paper such as ours could not fail to tackle, or at least touch on, a number of data issues, which may also form the object of future research. To mention one possible item for this agenda, the demand for and supply of funds at the group level could be investigated further and with it the governance of the intergroup liquidity redistribution. Daily data could be examined to zoom in on the market patterns along with a business case approach, e.g. from September to November 2008 when the process of liquidity distribution was especially impaired and certainly not only in Italy. Whether network topology is the right approach for this additional line of research depends on the data set used and the features (i) to (iv) we mentioned in the introduction of this paper could provide guidance. One would also very much like to broaden the analysis to an international basis, say encompassing the whole euro area interbank market. Unfortunately, right now there does not seem to be any suitable fully fledged data set, covering all bilateral exposures among banks resident in the area.

#### Measures of valued degree

This is a set of measures that mirrors those relating to the degree except that we consider the weight of the link. In this case it better serves the intuition of the resulting statistics to rely on relative measures, rather than absolute ones as in [3]-[5]:

[A.1]  $v_g^{OUT} = \frac{i_g^T W u}{u' W u} \in [0,1]$  valued out-degree of node g

[A.2] 
$$v_g^{IN} = \frac{i_g^{T} W^{T} u}{u' W u} \in [0,1]$$
 valued in-degree of node g

- [A.3]  $v_g \equiv \frac{i_g^T (W + W^T) u}{u^T W u} \in (0,1]$  valued degree of node g
- [A.4]  $\rho^{V} = \text{correlation}(v^{OUT}, v^{IN})$  valued degree correlation

where  $v^{OUT}$  and  $v^{IN}$  are  $n \times 1$  vectors which take on the g-th cell the value of  $v_g^{OUT}$  and  $v_{\sigma}^{IN}$  respectively.

**APPENDIX 2** 

#### **Measures of centrality**

A few notes on [10]: (i) the distance is 1 if the two nodes are linked directly, i.e. if  $b_{p,q} = 1$ ; (ii) the distance from p to q is the same as the one from q to p because it hinges on the symmetric matrix B; (iii) there may be more than one set of values (a', a", ...,  $a^{[m-1]}$ ) which links p and q, but (iv) there is no set formed by less than m-1 nodes, otherwise we would scale down  $d_{p,q}$  accordingly (this follows directly from the definition of distance as shortest path); (v) there may be no suitable set of a's, in which case the distance from p to q cannot be worked out or, alternatively, we say that such distance tends to infinite.

In parallel to the unweighted measure of the distance defined by [10], we can derive a weighted distance:

[A.5] 
$$d_{p,q}^{W} = d_{p,q} \times \left[\frac{4}{n} / \frac{(i_{p}^{T} + i_{q}^{T})(W + W^{T})(i_{p} + i_{q})}{u^{T}Wu}\right]$$

The loading factor between square parentheses is one (less than one/more than one) if the sum of the interbank deposits borrowed from and lent to both p and q is equal to (more/less than) 4/n times the sum of all interbank deposits binned in W. For example, the weighted distance is calculated shorter than its unweighted measure if p and q run relatively large amounts of interbank deposits.

A number of statistics can be derived from the matrices D and  $D^W$  which have as generic elements  $d_{p,q}$  and  $d_{p,q}^W$  respectively:

[A.6]	$h_g = \frac{i_g^{-1} D u}{n-1}$	average distance of node g
[A.7]	$\mathbf{h}_{g}^{W} = \frac{\mathbf{i}_{g}^{T} \mathbf{D}^{W} \mathbf{u}}{n-1}$	weighted average distance of node g
[A.8]	$h = \frac{u^{T} D u}{n(n-1)}$	average distance of the network
[A.9]	$\mathbf{h}^{\mathrm{W}} = \frac{\mathbf{u}^{\mathrm{T}} \mathbf{D}^{\mathrm{W}} \mathbf{u}}{n(n-1)}$	weighted average distance of the network
[A.10]	$e_{g} = max(i_{g}^{T}D)$	eccentricity of node g (maximum distance)
[A.11]	$e = max(e_g)$	eccentricity (diameter) of the network
[A.12]	$F_m = \frac{l(D=m)}{n(n-1)}$	mass distance function

 $h_g$  is the average distance from node g to the other n-1 nodes; h is the average across all g's;  $e_g$  is the maximum distance from bank g; F counts (in terms of relative frequency) the number of pairs whose distance is m $\in$  [1,n-1].  $h_g^W$ ,  $h^W$ are the weighted versions of  $h_g$  and h, while e is the aggregated measure of  $e_g$ . If the system is disconnected, [10] tends to  $\infty$  and measures [A.6]-[A.11] cannot be derived ([A.12] only up to a tail). Of course, the option remains to work them out on the resulting connected sub-sets established within the broader disconnected network. An alternative solution is to rely on harmonic measures, derived as functions of the inverse of the distance where the ratio between 1 and  $\infty$  is set equal to zero (of course, this is not strictly correct in the finite domain), so that pairs of nodes which are not linked give no contribution to the resulting sum (Newman, 2003).

**APPENDIX 3** 

#### **Measures of resilience**

To measure the shape of the split and, jointly, the size of the involved banks, our approach is (i) to identify the largest connected sub-network within the (n-1)-sized disconnected network; (ii) to derive as a residual the more isolated banks; (iii) to assess the weight of their deposits against the total deposits of the banks in the (n-1)-sized network. The formula we used is:

[A.13] 
$$r_{[-g]}^{W} = 1 - \frac{\left(u_{n-1}^{-g}\right)^{T} W^{-g} u_{n-1}^{-g}}{\left(u_{n-1}\right)^{T} W^{-g} u_{n-1}}$$

where  $u_{n-1}$  and  $u_{n-1}^{-g}$  are n-1 vectors, the former filled in with all 1's while the latter takes value 1 only corresponding to the banks belonging to the largest connected sub-network (zero elsewhere); W<sup>-g</sup> is derived from W, after having eliminated its g-th row and column.

To measure the geodesic frequency associated with each bank, we apply the following procedure. First, we compile all possible combinations of nodes p,q,g out of the *n* nodes. Then, for each triplet we verify whether any of the three nodes is geodesic to the other two. Once the check is run on all triplets, we compile a geodesic score for each node. Precisely, we are interested in the relative measure of this score, which we call the geodesic frequency. In algebraic terms and as regards node *g*, we use the formula:

[A.14]  $f_g = \frac{1(d_{p,q} = d_{p,g} + d_{g,q})}{[(n-1)(n-2)]/2}$  geodesic frequency of bank g

**APPENDIX 4** 

#### Additional checks on our dataset

#### Comparability of our dataset with other datasets studied in network topology

Figure A.1 shows a simple comparison of the network topology properties of our main dataset vis-à-vis other datasets examined in the specialized literature. On the x-axis we report the log size and on the y-axis the average path length. Data for the other networks are taken from Albert and Barabási (2002). Our dataset lies very close to the trend line and, based on this simple test, it would not seem to embody elements of peculiarity as far as its topology goes.



Data for the interbank market are a snapshot of September 2007. Data for other networks are from Albert and Barabási (2002).

#### Persistency in bilateral trading

In figure [A.2a] overleaf we set equal to 100 the number of bilateral deposits outstanding at the end of each month t and we show the number of those deposits that existed already at the end of month t-1. In figure [A2b], we present a measure of correlation between the deposits outstanding at the end of the two consecutive months, where this time the measure reflects not only where a given bilateral deposit existed but also its size. By and large, we observe a high

degree of persistency, with a correlation coefficient mostly above 0.8 in the 1990s (with a few downward spikes) and virtually equal to 1 since 2003. The finding of a high degree of persistency between consecutive month-end stocks should also suggest that had comparable intra-month observations been available, they would not have painted a very different picture.



*Changes in population* 

Over the almost 20 years covered by our sample, there were a number of changes in the population of banks resident in Italy. As a result, the patterns followed by a number of our statistics may, in principle, owe to a mixture of developments in the underlying economic phenomenon and to population changes.

This is a common and well-known problem in statistics, typically dealt with by introducing the concept of "flow" next to that of "difference in stocks". The first differs from the second precisely because "flows" adjust for population changes (other adjustments are also common, e.g. the impact of exchange rate changes in the valuation of stocks, but are not relevant in our case). The next step is the derivation of an index series based at the start of the sample. The procedure is described in more detail on the first page of the technical notes in the Statistics Section of the ECB monthly bulletin. We show below the results of an exercise along these lines on the total amount of interbank deposits outstanding at end June from 1990 to 2008. For ease of comparison, stocks are rebased to 100 in June 1990. Until 2002, changes in the population had virtually no impact. Thereafter some gap opened, of which roughly half can be attributed to one

observation (June 2003). In any case the slope of the series and its smoothness do not seem to be affected.



#### REFERENCES

- Albert R. and A.-L. Barabási (2002), "Statistical mechanics of complex networks". <u>Review of modern physics</u>, 74, pp. 47-97.
- Albert R., H. Jeong and A.-L. Barabási (1999), "Diameter of the world-wide web". Nature, 401, pp. 130-131.
- Albert R., H. Jeong and A.-L. Barabási (2000), "Error and attack tolerance of complex networks". <u>Nature</u>, 406, pp. 378-382.
- Allen F. and D. Gale (2000), "Financial Contagion". Journal of Political Economy, 108:1, pp. 1-33.
- Angelini P., G. Maresca and D. Russo (1996), "Systemic risk in the netting system", Journal of Banking and Finance, 20, pp. 853-868.
- Bech M.L. and E. Atalay (2008), "The Topology of the Federal Funds Market", <u>Federal</u> <u>Reserve Bank of New York Staff Reports</u>, no. 354.
- Becher C., S. Millard and K. Soramäli (2008), "The network topology of CHAPS Sterling", <u>Bank of England Working Paper</u>, no. 355.
- Bernanke B.S. (2008), "Reducing systemic risk", speech at the Federal Reserve Bank of Kansas City's Annual Economic Symposium, 22 August.
- BIS (2006), Cross-border collateral arrangements, CPSS Publications, no 71.
- Blåvarg M. and P. Nimander (2002), "Inter-bank exposures and systemic risk". Sveriges Riksbank, Economic Review, pp. 19-45.
- Bollobás B. (1998), Modern Graph Theory, Springer, New York.
- Boss M., H. Elsinger, M. Summers and S. Thurner (2004), "The Network Topology of the Interbank Market". <u>Quantitative Finance</u>, 4:6, pp. 677-684.
- Brusco S. and F. Castiglionesi (2007), "Liquidity Coinsurance, Moral Hazard and Financial Contagion". <u>The Journal of Finance</u>, 62:5, pp. 2275-2302.
- Bryant J. (1980), "A Model of Reserves, Bank Runs, and Deposit Insurance". Journal of Banking and Finance, 4, pp. 335-344.
- Cajueiro D.O. and B.M Tabak (2007), "The role of banks in the Brazilian Interbank Market: Does bank type matter?", <u>Banco Central do Brasil Working Paper</u>, no. 130.
- Cifuentes R. (2004), "Banking Concentration: Implications for Systemic Risk and Safety-Net Design", in <u>Banking Market Structure and Monetary Policy</u>, ed. L. Ahumada and J. Fuentes, Central Bank of Chile.
- Degryse H. and G. Nguyen (2007), "Interbank exposures: An empirical examination of Contagion Risk in the Belgian Banking System". <u>International Journal of Central Banking</u>, June, pp. 123-171.
- Devriese J. and J. Mitchell (2005), "Liquidity risk in securities settlement". <u>National</u> <u>Bank of Belgium Working Paper Research</u>, no. 72.
- Diamond D.W. and P.H. Dybvig (1983), "Bank Runs, Deposit Insurance, and Liquidity". Journal of Political Economy, 91, pp. 401-419.
- Donaldson R.G. (1992), "Costly Liquidation, Interbank Trade, Bank Runs and Panics". Journal of financial intermediation, 2, pp. 59-82.
- Eisenberg L. and T. Noe (2001), "Systemic Risk in Financial Systems". <u>Management</u> <u>Science</u>, 47:2, pp. 236-249.
- Freixas X., B. Parigi and J.-C. Rochet (2000), "Systemic risk, interbank relations and liquidity provision by the central bank". Journal of Money, Credit and Banking, 35:1, pp. 111-128.
- Furfine C.H. (2003), "Interbank Exposures: Quantifying the Risk of Contagion". Journal of Money, Credit, and Banking, 35:1, pp. 111-128.
- Iori G, G. Caldarelli, G. Gabbi, G. de Masi and O.V. Precup (2007), "A Network Analysis of the Italian Overnight Money Market", Journal of Economic Dynamics and Control, 32, pp. 259-278.
- Iyer R. and J.L. Peydró-Alcalde (2005), "Interbank contagion: Evidence from real transactions". In ECB, Risk measurement and systemic risk.

- Jacklin C.J. and S. Bhattacharya (1988), "Distinguishing Panics and Information-Based Bank Runs: Welfare and Policy Implications". Journal of Political Economy, 96:3, pp. 568-592.
- Kindleberger, C.P. (1978), <u>Manias, Panics, and Crashes: A History of Financial Crises</u>, New York: Basic Books.
- Koponen R. and K. Soramäki (1998), "Intraday liquidity needs in a modern interbank payment system. A simulation approach". <u>Bank of Finland Studies</u>, no. 14-1998.
- van Lelyveld I. and F. Liedorp (2006), "Interbank Contagion in the Dutch Banking Sector: A Sensitivity Analysis". <u>International Journal of Central Banking</u>, 2:2, pp. 99-133.
- Manna M. (2004), "Developing statistical indicators of the integration of the euro area banking system". <u>ECB Working Paper Series</u>, no. 300.

Milgram S. (1967), "The Small World Problem", Psychology Today, 2, pp. 60-67.

- Mistrulli P.E. (2007), "Assessing financial contagion in the interbank market: Maximum entropy versus observed interbank lending patterns". <u>Banca d'Italia</u> <u>Temi di discussione (Working Papers)</u>, no. 641.
- Müller J. (2003), "Two Approaches to Assess Contagion in the Interbank Market". Mimeo, Swiss National Bank.
- Müller J. (2006), "Interbank Credit Lines as a Channel of Contagion". Journal of <u>Financial Services Research</u>, 29:1, pp. 37-60.
- Newman M.E.J. (2003), "The structure and Function of Complex Network". <u>Society</u> for Industrial and. Applied Mathematic (SIAM) Review, 45:2, pp. 167-256.
- Nier E., J. Yang, T. Yorulmazer and A. Alentorn (2008), "Network models and financial stability". <u>Bank of England working paper</u>, no. 346.
- von Peter G. (2007), "International banking centres: a network perspective". <u>BIS</u> <u>Quarterly Review</u>, December, pp. 33-45.
- Rochet J.-C. and J. Tirole (1996) "Interbank Lending and Systemic Risk". Journal of Money, Credit and Banking, 28:4, pp. 733-762.
- Sheldon G. and M. Maurer (1998), "Interbank lending and systemic risk". <u>Swiss</u> Journal of Economics and Statistics, 134:4.2, pp. 684-704.
- Soramäki K., Arnold J., M.L. Bech, W.E. Beyeler and R.J. Glass (2006), "The Topology of Interbank Payment Flows". <u>Federal Reserve Bank of New York Staff Reports</u>, no. 243.
- Upper C. and A. Worms (2004), "Estimating Bilateral Exposures in the German Interbank Market: Is There a Danger of Contagion?", <u>European Economic Review</u>, 48:4, pp. 827-849.

Watts D.J. (2003), Six Degrees: The Science of a Connected Age. Norton, New York.

- Watts D.J. and S.H. Strogatz (1998), "Collective dynamics of small world networks". <u>Nature</u>, 393, pp. 440-442.
- Wells S. (2002), "UK interbank exposures: systemic risk implications". <u>Bank of England, Financial Stability Review</u>, December.

#### RECENTLY PUBLISHED "TEMI" (\*)

- N. 684 Balancing work and family in Italy: New mothers' employment decisions after childbirth, by Piero Casadio, Martina Lo Conte and Andrea Neri (August 2008).
- N. 685 *Temporal aggregation of univariate and multivariate time series models: A survey*, by Andrea Silvestrini and David Veredas (August 2008).
- N. 686 Exploring agent-based methods for the analysis of payment systems: A crisis model for StarLogo TNG, by Luca Arciero, Claudia Biancotti, Leandro D'Aurizio and Claudio Impenna (August 2008).
- N. 687 The labor market impact of immigration in Western Germany in the 1990's, by Francesco D'Amuri, Gianmarco I. P. Ottaviano and Giovanni Peri (August 2008).
- N. 688 Agglomeration and growth: the effects of commuting costs, by Antonio Accetturo (September 2008).
- N. 689 A beta based framework for (lower) bond risk premia, by Stefano Nobili and Gerardo Palazzo (September 2008).
- N. 690 *Nonlinearities in the dynamics of the euro area demand for M1*, by Alessandro Calza and Andrea Zaghini (September 2008).
- N. 691 Educational choices and the selection process before and after compulsory schooling, by Sauro Mocetti (September 2008).
- N. 692 Investors' risk attitude and risky behavior: a Bayesian approach with imperfect information, by Stefano Iezzi (September 2008).
- N. 693 Competing influence, by Enrico Sette (September 2008).
- N. 694 *La relazione tra gettito tributario e quadro macroeconomico in Italia*, by Alberto Locarno and Alessandra Staderini (December 2008).
- N. 695 *Immigrant earnings in the Italian labour market*, by Antonio Accetturo and Luigi Infante (December 2008).
- N. 696 *Reservation wages: explaining some puzzling regional patterns*, by Paolo Sestito and Eliana Viviano (December 2008).
- N. 697 Technological change and the demand for currency: an analysis with household data, by Francesco Lippi and Alessandro Secchi (December 2008).
- N. 698 Immigration and crime: an empirical analysis, by Milo Bianchi, Paolo Buonanno and Paolo Pinotti (December 2008).
- N. 699 Bond risk premia, macroeconomic fundamentals and the exchange rate, by Marcello Pericoli and Marco Taboga (January 2009).
- N. 700 What determines debt intolerance? The role of political and monetary institutions, by Raffaela Giordano and Pietro Tommasino (January 2009).
- N. 701 On analysing the world distribution of income, by Anthony B. Atkinson and Andrea Brandolini (January 2009).
- N. 702 Dropping the books and working off the books, by Rita Cappariello and Roberta Zizza (January 2009).
- N. 703 Measuring wealth mobility, by Andrea Neri (January 2009).
- N. 704 Oil and the macroeconomy: a quantitative structural analysis, by Francesco Lippi and Andra Nobili (March 2009).
- N. 705 *The (mis)specification of discrete duration models with unobserved heterogeneity: a Monte Carlo study*, by Cheti Nicoletti and Concetta Rondinelli (March 2009).
- N. 706 *Macroeconomic effects of grater competition in the service sector: the case of Italy*, by Lorenzo Forni, Andrea Gerali and Massimiliano Pisani (March 2009).
- N. 707 What determines the size of bank loans in industrialized countries? The role of government debt, by Riccardo De Bonis and Massimiliano Stacchini (March 2009).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

# F. BUSETTI, Tests of seasonal integration and cointegration in multivariate unobserved component models, Journal of Applied Econometrics, Vol. 21, 4, pp. 419-438, **TD No. 476 (June 2003).**

- C. BIANCOTTI, A polarization of inequality? The distribution of national Gini coefficients 1970-1996, Journal of Economic Inequality, Vol. 4, 1, pp. 1-32, **TD No. 487 (March 2004).**
- L. CANNARI and S. CHIRI, La bilancia dei pagamenti di parte corrente Nord-Sud (1998-2000), in L. Cannari, F. Panetta (a cura di), Il sistema finanziario e il Mezzogiorno: squilibri strutturali e divari finanziari, Bari, Cacucci, TD No. 490 (March 2004).
- M. BOFONDI and G. GOBBI, *Information barriers to entry into credit markets*, Review of Finance, Vol. 10, 1, pp. 39-67, **TD No. 509 (July 2004).**
- W. FUCHS and LIPPI F., *Monetary union with voluntary participation*, Review of Economic Studies, Vol. 73, pp. 437-457 **TD No. 512** (July 2004).
- E. GAIOTTI and A. SECCHI, *Is there a cost channel of monetary transmission? An investigation into the pricing behaviour of 2000 firms*, Journal of Money, Credit and Banking, Vol. 38, 8, pp. 2013-2038
  TD No. 525 (December 2004).
- A. BRANDOLINI, P. CIPOLLONE and E. VIVIANO, *Does the ILO definition capture all unemployment?*, Journal of the European Economic Association, Vol. 4, 1, pp. 153-179, **TD No. 529** (December 2004).
- A. BRANDOLINI, L. CANNARI, G. D'ALESSIO and I. FAIELLA, *Household wealth distribution in Italy in the* 1990s, in E. N. Wolff (ed.) International Perspectives on Household Wealth, Cheltenham, Edward Elgar, **TD No. 530 (December 2004).**
- P. DEL GIOVANE and R. SABBATINI, Perceived and measured inflation after the launch of the Euro: Explaining the gap in Italy, Giornale degli economisti e annali di economia, Vol. 65, 2, pp. 155-192, TD No. 532 (December 2004).
- M. CARUSO, *Monetary policy impulses, local output and the transmission mechanism*, Giornale degli economisti e annali di economia, Vol. 65, 1, pp. 1-30, **TD No. 537 (December 2004).**
- L. GUISO and M. PAIELLA, The role of risk aversion in predicting individual behavior, In P. A. Chiappori e C. Gollier (eds.) Competitive Failures in Insurance Markets: Theory and Policy Implications, Monaco, CESifo, **TD No. 546 (February 2005).**
- G. M. TOMAT, Prices product differentiation and quality measurement: A comparison between hedonic and matched model methods, Research in Economics, Vol. 60, 1, pp. 54-68, TD No. 547 (February 2005).
- L. GUISO, M. PAIELLA and I. VISCO, Do capital gains affect consumption? Estimates of wealth effects from Italian household's behavior, in L. Klein (ed), Long Run Growth and Short Run Stabilization: Essays in Memory of Albert Ando (1929-2002), Cheltenham, Elgar, TD No. 555 (June 2005).
- F. BUSETTI, S. FABIANI and A. HARVEY, *Convergence of prices and rates of inflation*, Oxford Bulletin of Economics and Statistics, Vol. 68, 1, pp. 863-878, **TD No. 575 (February 2006).**
- M. CARUSO, Stock market fluctuations and money demand in Italy, 1913 2003, Economic Notes, Vol. 35, 1, pp. 1-47, **TD No. 576 (February 2006).**
- R. BRONZINI and G. DE BLASIO, *Evaluating the impact of investment incentives: The case of Italy's Law* 488/92. Journal of Urban Economics, Vol. 60, 2, pp. 327-349, **TD No. 582 (March 2006).**
- R. BRONZINI and G. DE BLASIO, Una valutazione degli incentivi pubblici agli investimenti, Rivista Italiana degli Economisti, Vol. 11, 3, pp. 331-362, **TD No. 582** (March 2006).
- A. DI CESARE, *Do market-based indicators anticipate rating agencies? Evidence for international banks*, Economic Notes, Vol. 35, pp. 121-150, **TD No. 593 (May 2006).**
- R. GOLINELLI and S. MOMIGLIANO, *Real-time determinants of fiscal policies in the euro area*, Journal of Policy Modeling, Vol. 28, 9, pp. 943-964, **TD No. 609 (December 2006).**

#### 2006

- S. SIVIERO and D. TERLIZZESE, *Macroeconomic forecasting: Debunking a few old wives' tales*, Journal of Business Cycle Measurement and Analysis, v. 3, 3, pp. 287-316, **TD No. 395 (February 2001).**
- S. MAGRI, Italian households' debt: The participation to the debt market and the size of the loan, Empirical Economics, v. 33, 3, pp. 401-426, **TD No. 454 (October 2002).**
- L. CASOLARO. and G. GOBBI, *Information technology and productivity changes in the banking industry*, Economic Notes, Vol. 36, 1, pp. 43-76, **TD No. 489 (March 2004).**
- G. FERRERO, *Monetary policy, learning and the speed of convergence,* Journal of Economic Dynamics and Control, v. 31, 9, pp. 3006-3041, **TD No. 499 (June 2004).**
- M. PAIELLA, Does wealth affect consumption? Evidence for Italy, Journal of Macroeconomics, Vol. 29, 1, pp. 189-205, TD No. 510 (July 2004).
- F. LIPPI. and S. NERI, *Information variables for monetary policy in a small structural model of the euro area*, Journal of Monetary Economics, Vol. 54, 4, pp. 1256-1270, **TD No. 511 (July 2004).**
- A. ANZUINI and A. LEVY, *Monetary policy shocks in the new EU members: A VAR approach*, Applied Economics, Vol. 39, 9, pp. 1147-1161, **TD No. 514 (July 2004).**
- D. JR. MARCHETTI and F. Nucci, *Pricing behavior and the response of hours to productivity shocks*, Journal of Money Credit and Banking, v. 39, 7, pp. 1587-1611, **TD No. 524 (December 2004).**
- R. BRONZINI, *FDI Inflows, agglomeration and host country firms' size: Evidence from Italy*, Regional Studies, Vol. 41, 7, pp. 963-978, **TD No. 526 (December 2004).**
- L. MONTEFORTE, Aggregation bias in macro models: Does it matter for the euro area?, Economic Modelling, 24, pp. 236-261, **TD No. 534 (December 2004).**
- A. NOBILI, Assessing the predictive power of financial spreads in the euro area: does parameters instability matter?, Empirical Economics, Vol. 31, 1, pp. 177-195, **TD No. 544 (February 2005).**
- A. DALMAZZO and G. DE BLASIO, *Production and consumption externalities of human capital: An empirical study for Italy*, Journal of Population Economics, Vol. 20, 2, pp. 359-382, **TD No. 554 (June 2005).**
- M. BUGAMELLI and R. TEDESCHI, Le strategie di prezzo delle imprese esportatrici italiane, Politica Economica, v. 23, 3, pp. 321-350, TD No. 563 (November 2005).
- L. GAMBACORTA and S. IANNOTTI, Are there asymmetries in the response of bank interest rates to monetary shocks?, Applied Economics, v. 39, 19, pp. 2503-2517, TD No. 566 (November 2005).
- P. ANGELINI and F. LIPPI, *Did prices really soar after the euro cash changeover? Evidence from ATM withdrawals*, International Journal of Central Banking, Vol. 3, 4, pp. 1-22, **TD No. 581 (March 2006).**
- A. LOCARNO, *Imperfect knowledge, adaptive learning and the bias against activist monetary policies*, International Journal of Central Banking, v. 3, 3, pp. 47-85, **TD No. 590 (May 2006).**
- F. LOTTI and J. MARCUCCI, *Revisiting the empirical evidence on firms' money demand*, Journal of Economics and Business, Vol. 59, 1, pp. 51-73, **TD No. 595** (May 2006).
- P. CIPOLLONE and A. ROSOLIA, *Social interactions in high school: Lessons from an earthquake*, American Economic Review, Vol. 97, 3, pp. 948-965, **TD No. 596 (September 2006).**
- L. DEDOLA and S. NERI, *What does a technology shock do? A VAR analysis with model-based sign restrictions*, Journal of Monetary Economics, Vol. 54, 2, pp. 512-549, **TD No. 607 (December 2006).**
- F. VERGARA CAFFARELLI, *Merge and compete: strategic incentives for vertical integration*, Rivista di politica economica, v. 97, 9-10, serie 3, pp. 203-243, **TD No. 608 (December 2006).**
- A. BRANDOLINI, Measurement of income distribution in supranational entities: The case of the European Union, in S. P. Jenkins e J. Micklewright (eds.), Inequality and Poverty Re-examined, Oxford, Oxford University Press, TD No. 623 (April 2007).
- M. PAIELLA, *The foregone gains of incomplete portfolios*, Review of Financial Studies, Vol. 20, 5, pp. 1623-1646, **TD No. 625 (April 2007).**
- K. BEHRENS, A. R. LAMORGESE, G.I.P. OTTAVIANO and T. TABUCHI, *Changes in transport and non transport costs: local vs. global impacts in a spatial network*, Regional Science and Urban Economics, Vol. 37, 6, pp. 625-648, **TD No. 628** (April 2007).
- M. BUGAMELLI, Prezzi delle esportazioni, qualità dei prodotti e caratteristiche di impresa: analisi su un campione di imprese italiane, v. 34, 3, pp. 71-103, Economia e Politica Industriale, TD No. 634 (June 2007).
- G. ASCARI and T. ROPELE, *Optimal monetary policy under low trend inflation*, Journal of Monetary Economics, v. 54, 8, pp. 2568-2583, **TD No. 647** (November 2007).

- R. GIORDANO, S. MOMIGLIANO, S. NERI and R. PEROTTI, *The Effects of Fiscal Policy in Italy: Evidence from a VAR Model*, European Journal of Political Economy, Vol. 23, 3, pp. 707-733, **TD No. 656** (January 2008).
- B. ROFFIA and A. ZAGHINI, *Excess money growth and inflation dynamics*, International Finance, v. 10, 3, pp. 241-280, **TD No. 657 (January 2008).**
- G. BARBIERI, P. CIPOLLONE and P. SESTITO, Labour market for teachers: demographic characteristics and allocative mechanisms, Giornale degli economisti e annali di economia, v. 66, 3, pp. 335-373, TD No. 672 (June 2008).
- E. BREDA, R. CAPPARIELLO and R. ZIZZA, Vertical specialisation in Europe: evidence from the import content of exports, Rivista di politica economica, numero monografico, TD No. 682 (August 2008).

2008

- P. ANGELINI, *Liquidity and announcement effects in the euro area*, Giornale degli Economisti e Annali di Economia, v. 67, 1, pp. 1-20, **TD No. 451 (October 2002).**
- P. ANGELINI, P. DEL GIOVANE, S. SIVIERO and D. TERLIZZESE, Monetary policy in a monetary union: What role for regional information?, International Journal of Central Banking, v. 4, 3, pp. 1-28, TD No. 457 (December 2002).
- F. SCHIVARDI and R. TORRINI, *Identifying the effects of firing restrictions through size-contingent Differences in regulation*, Labour Economics, v. 15, 3, pp. 482-511, **TD No. 504 (June 2004).**
- L. GUISO and M. PAIELLA,, *Risk aversion, wealth and background risk*, Journal of the European Economic Association, v. 6, 6, pp. 1109-1150, **TD No. 483 (September 2003).**
- C. BIANCOTTI, G. D'ALESSIO and A. NERI, *Measurement errors in the Bank of Italy's survey of household income and wealth*, Review of Income and Wealth, v. 54, 3, pp. 466-493, **TD No. 520 (October 2004).**
- S. MOMIGLIANO, J. HENRY and P. HERNÁNDEZ DE COS, The impact of government budget on prices: Evidence from macroeconometric models, Journal of Policy Modelling, v. 30, 1, pp. 123-143 TD No. 523 (October 2004).
- L. GAMBACORTA, *How do banks set interest rates?*, European Economic Review, v. 52, 5, pp. 792-819, **TD No. 542 (February 2005).**
- P. ANGELINI and A. GENERALE, On the evolution of firm size distributions, American Economic Review, v. 98, 1, pp. 426-438, **TD No. 549 (June 2005).**
- R. FELICI and M. PAGNINI, *Distance, bank heterogeneity and entry in local banking markets*, The Journal of Industrial Economics, v. 56, 3, pp. 500-534, **No. 557 (June 2005).**
- S. DI ADDARIO and E. PATACCHINI, *Wages and the city. Evidence from Italy*, Labour Economics, v.15, 5, pp. 1040-1061, **TD No. 570 (January 2006).**
- M. PERICOLI and M. TABOGA, Canonical term-structure models with observable factors and the dynamics of bond risk premia, Journal of Money, Credit and Banking, v. 40, 7, pp. 1471-88, TD No. 580 (February 2006).
- E. VIVIANO, Entry regulations and labour market outcomes. Evidence from the Italian retail trade sector, Labour Economics, v. 15, 6, pp. 1200-1222, **TD No. 594 (May 2006).**
- S. FEDERICO and G. A. MINERVA, Outward FDI and local employment growth in Italy, Review of World Economics, Journal of Money, Credit and Banking, v. 144, 2, pp. 295-324, TD No. 613 (February 2007).
- F. BUSETTI and A. HARVEY, *Testing for trend*, Econometric Theory, v. 24, 1, pp. 72-87, **TD No. 614** (February 2007).
- V. CESTARI, P. DEL GIOVANE and C. ROSSI-ARNAUD, *Memory for prices and the Euro cash changeover: an analysis for cinema prices in Italy*, In P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, **TD No. 619 (February 2007)**.
- B. H. HALL, F. LOTTI and J. MAIRESSE, Employment, innovation and productivity: evidence from Italian manufacturing microdata, Industrial and Corporate Change, v. 17, 4, pp. 813-839, TD No. 622 (April 2007).

- J. SOUSA and A. ZAGHINI, *Monetary policy shocks in the Euro Area and global liquidity spillovers,* International Journal of Finance and Economics, v.13, 3, pp. 205-218, **TD No. 629 (June 2007).**
- M. DEL GATTO, GIANMARCO I. P. OTTAVIANO and M. PAGNINI, Openness to trade and industry cost dispersion: Evidence from a panel of Italian firms, Journal of Regional Science, v. 48, 1, pp. 97-129, TD No. 635 (June 2007).
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "inflation perceptions"? A survey-based analysis of Italian consumers, in P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 655 (January 2008).
- B. BORTOLOTTI, and P. PINOTTI, *Delayed privatization*, Public Choice, v. 136, 3-4, pp. 331-351, TD No. 663 (April 2008).
- R. BONCI and F. COLUMBA, *Monetary policy effects: New evidence from the Italian flow of funds*, Applied Economics, v. 40, 21, pp. 2803-2818, **TD No. 678 (June 2008).**
- M. CUCCULELLI, and G. MICUCCI, *Family Succession and firm performance: evidence from Italian family firms*, Journal of Corporate Finance, v. 14, 1, pp. 17-31, **TD No. 680 (June 2008).**
- A. SILVESTRINI and D. VEREDAS, *Temporal aggregation of univariate and multivariate time series models: a survey*, Journal of Economic Surveys, v. 22, 3, pp. 458-497, **TD No. 685 (August 2008).**

2009

- S. MAGRI, *The financing of small innovative firms: the Italian case*, Economics of Innovation and New Technology, v. 18, 2, pp. 181-204, **TD No. 640 (September 2007).**
- S. MAGRI, *The financing of small entrepreneurs in Italy*, Annals of Finance, v. 5, 3-4, pp. 397-419, **TD** No. 640 (September 2007).
- R. GOLINELLI and S. MOMIGLIANO, *The Cyclical Reaction of Fiscal Policies in the Euro Area. A Critical Survey of Empirical Research*, Fiscal Studies, v. 30, 1, pp. 39-72, **TD No. 654 (January 2008).**
- L. ARCIERO, C. BIANCOTTI, L. D'AURIZIO and C. IMPENNA, *Exploring agent-based methods for the analysis* of payment systems: A crisis model for StarLogo TNG, Journal of Artificial Societies and Social Simulation, v. 12, 1, **TD No. 686 (August 2008).**
- A. CALZA and A. ZAGHINI, Nonlinearities in the dynamics of the euro area demand for M1, Macroeconomic Dynamics, v. 13, 1, pp. 1-19, **TD No. 690 (September 2008).**
- L. FRANCESCO and A. SECCHI, *Technological change and the households' demand for currency*, Journal of Monetary Economics, v. 56, 2, pp. 222-230, **TD No. 697 (December 2008).**

#### FORTHCOMING

- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, **TD No. 458 (December 2002).**
- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD** No. 578 (February 2006).
- M. PERICOLI and M. SBRACIA, *The CAPM and the risk appetite index: theoretical differences, empirical similarities, and implementation problems*, International Finance, **TD No. 586 (March 2006).**
- R. BRONZINI and P. PISELLI, *Determinants of long-run regional productivity with geographical spillovers: the role of R&D, human capital and public infrastructure,* Regional Science and Urban Economics, **TD No. 597 (September 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, **TD No. 598 (September 2006).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and the business cycle*, Journal of Financial Stability, **TD No. 601 (September 2006).**
- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, **TD No. 637 (June 2007).**

- V. DI GIACINTO and G. MICUCCI, *The producer service sector in Italy: long-term growth and its local determinants*, Spatial Economic Analysis, **TD No. 643 (September 2007).**
- L. FORNI, L. MONTEFORTE and L. SESSA, *The general equilibrium effects of fiscal policy: Estimates for the euro area,* Journal of Public Economics, **TD No. 652 (November 2007).**
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS, *Securitisation and the bank lending channel*, European Economic Review, **TD No. 653 (November 2007).**
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, *What's behind "Inflation Perceptions"? A survey-based analysis of Italian consumers*, Giornale degli Economisti e Annali di Economia, **TD No. 655** (January 2008).
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, **TD No. 688 (September 2008).**
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, **TD No. 706 (March 2009).**